As man y of y ou noticed, the b yteco des that w ere giv en to y ou for Assignmen t #5 w ere not su�cien t

to write a binary searc h program.

Y ou need 2 more b yteco des, namely IALO

AD and IASTORE to load/store to/from the stac k in a

lo catio n in lo cal storage whose address is computed rather than a constan t.

More precisely;

IALO

AD (b yteco de 0x2e) tak es the top of the stac k as an index in the lo cal storage area and loads

the con ten ts of that lo cal storage w ord on top of the stac k (destro ying the index v alue).

That is if the lo cal storage area lo oks lik e (the left column is the index, the righ t column is the

con ten ts of the lo cal storage area):

----------

0 17

----------

1 73

----------

2 4

----------

3 27

----------

4 13

----------

etc.

and if the stac k lo oks lik e

----------

19 <---- "below" top of the stack

----------

3 <---- top of the stack

----------

The b yteco de IALO

AD will result in the new stac k con�guration

----------

19

----------

27 <---- top of the stack

----------

IASTORE (b yteco de 0x4f ) stores the elemen t b elo w the top of the stac k at the lo cal storage lo catio n

whose index is on the top of the stac k. Then the t w o top en tries on the stac k are p opp ed.

That is with the initial conditions:

----------

0 17

----------

1 73

----------

2 4

----------

3 27

----------

4 13

----------

etc.

and

----------

214

----------

19 <---- "below" top of the stack

----------

3 <---- top of the stack

----------

IASTORE w ould result in

----------

0 17

----------

1 73

----------

2 4

----------

3 19

----------

4 13

----------

etc.

and

----------

214 <---- top of the stack

----------

Thanks to all of y ou who noticed this problem.

Note that y ou can create y our o wn lo cal v ariables. If "k ey" is at lo cati on 0 (in lo cal storage) , "size"

at lo cati on 1, and "arra y" at the next "size" lo cati ons, the lo catio ns "size + 2", "size + 3" etc can

b e used to store temp orary v ariables, if need b e.

In the "real" JVM IALO

AD and IASTORE are sligh tly more complex b ecause the lo cal storage

area is more complex. Y ou ha v e to iden tify the "ob ject" (e.g., arra y) that y ou w an t to index.

JLB and Saurabh